Author's response to reviews

Title: Ureteral reconstruction using a tapered non-vascularized bladder graft: an experimental study in a canine animal model

Authors:
Lujia Zou (zlj_880522@163.com)
Shanhua Mao (maoshanhua@163.com)
Shenghua Liu (08301010098@fudan.edu.cn)
Limin Zhang (alexzl1126@126.com)
Tian Yang (09301010090@fudan.edu.cn)
Yun Hu (10301010087@fudan.edu.cn)
Qiang Ding (huashanurologyqd@163.com)
Haowen Jiang (huashanurologyhwj@163.com)

Version: 3 Date: 09 Oct 2017

Author’s response to reviews:
Response Letter for Minor Revision

1) E-mail addresses:

The email addresses for all authors have been checked consistent between the title page and the submission system. The email address of a co-author has been modified in the title page.

2) Copy edit:

All of the recommended amendments to the wording have been followed.